\documentclass[12pt]{amsart}
\usepackage{amssymb}
\usepackage{enumitem}
\usepackage[paperheight=11in,paperwidth=8.5in,left=1in,top=1in,right=1in,bottom=1in]{geometry}
\usepackage{hyperref}
\usepackage{mathtools}

\newcommand{\prob}{\mathbb{P}}
\newcommand{\E}{\mathbb{E}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\R}{\mathbb{R}}
\newcommand{\pr}{\mathbb{P}}
\renewcommand{\thefootnote}{\fnsymbol{footnote}}
\newcommand{\op}[1]{\operatorname{#1}}
\newcommand{\mcal}[1]{\mathcal{#1}}

\begin{document}
\title{Stat 150 Extra Credit}
\author{Fall 2021}
\maketitle

\textbf{Due Friday, November 19th, at 11:59 PM on Gradescope.}\\

\bigskip{}

A maximum of 5 points to the second midterm will be awarded on the basis of this assignment. Your score on the second midterm is capped at a 65/65. For example, if you scored a 62.5/65, then this assignment can bring your score to a maximum of 65/65.

\begin{enumerate}[label=\arabic*.]\setlength{\itemsep}{10pt}

\item Look through your second midterm on Gradescope carefully.

\item Look through the solutions to the second midterm on bCourses carefully.

\item For each problem, explain your thought process and how it compared to the solutions. Two to three sentences would be fine, but you are welcome to write more. If you got a perfect score on a particular problem, then your explanation for that problem can be brief.

\item For each problem, explain why you found it difficult (or easy). Again, two to three sentences would be fine, but you are welcome to write more. As before, if you got a perfect score on a problem, you can be brief.
    
\end{enumerate}

\end{document}